**Author’s response to reviews**

**Title:** Attitudes toward Integrative Pediatrics: a National Survey among Youth Health Care Physicians in the Netherlands

**Authors:**

- Miek C Jong (m.jong@louisbolk.nl)
- Marja Van Vliet (m.vanvliet@louisbolk.nl)
- Susan Huttenhuis (susanhuttenhuis@gmail.com)
- Désirée van der Veer (d.vanderveer@online.nl)
- Silvia van den Heijkant (s.vandenheijkant@vumc.nl)

**Version:** 4  **Date:** 7 December 2011

**Author’s response to reviews:** see over
Driebergen, 7 December 2011

BMC Central Editorial
c/o Editor in chief

Concerning: Submission of revised manuscript MS 6541363395787391

Dear Editor,

Thank you so much for acceptance of our manuscript in your journal. Hereby we would like to resubmit the revised lay-out of the manuscript.

"Attitudes toward Integrative Pediatrics: a National Survey among Youth Health Care Physicians in the Netherlands"

According to the comments of the editorial office, we have tackled the lay-out issues.

Looking forward to its publication in the BMC CAM journal.

With kind regards, on behalf of all authors

dr. Miek Jong
Managing Director

Louis Bolk Institute
Hoofdstraat 24
3972 LA Driebergen
The Netherlands
Ph: +31 343 523 860
Email: m.jong@louisbolk.nl